Author's response to reviews

Title: Osteomyelitis and pyoarthritis caused by localized paracoccidioidomycosis in an immunocompetent patient: a case report

Authors:

Michel S Michelan (michelsmichelan@yahoo.com.br)
Eloy A Fernandes (fernandes_ea@yahoo.com.br)
Leonardo F Freitas (furtadoleofreitas@hotmail.com)
Rodrigo H Ribeiro (centrodeestudosiamspe@yahoo.com.br)
Marilia M Milano (michelsmichelan@yahoo.com.br)
Soraya S Monteiro (sorayasmonteiro@bol.com.br)

Version: 3 Date: 25 June 2012

Author's response to reviews:

Dear reviewers,

We perform the requested modifications, as follows:
- Ethnicity of the patient;
- We mention the phrase "Furthermore, isolated bone involvement is very rare, especially in children," according to the literature;
- Dose of fluconazole and continued therapy;
- Details regarding the treatment of disease;
- Better information about the clinic and epidemiology of the disease;
- Add the radiographs of the chest and pelvis.
- To the description of the condition (blade), we chose to keep it in the legend to Figure not to further increase the text;
- Reduce the abstract to less than 300 words.

We are available,
Thank you,
Leonardo Furtado Freitas